Author's response to reviews

Title: Nicotine patches and quitline counseling to help hospitalized smokers stay quit: Study protocol for a randomized controlled trial

Authors:

Sharon Cummins (scummins@ucsd.edu)
Shu-Hong Zhu (szhu@ucsd.edu)
Anthony Gamst (acgamst@ucsd.edu)
Carrie Kirby (ckirby@ucsd.edu)
Kendra Brandstein (Brandstein.Kendra@scrippshealth.org)
Hillary Klonhoff-Cohen (hklonoffcohen@ucsd.edu)
Edward Chaplin (Chaplin.Edward@scrippshealth.org)
Timothy Morris (T1morris@ucsd.edu)
Gregory Seymann (gseymann@ucsd.edu)
Joshua Lee (Jol002@ucsd.edu)

Version: 4 Date: 24 January 2012

Author's response to reviews:

We have changed the title per the editor's request.